Author's response to reviews

Title: Predictors of web-based follow-up response in the Prevention of Low Back Pain the Military Trial

Authors:

John D Childs (john@texpts.com)
Deydre S Teyhen (dteyhen@sbcglobal.net)
Joshua J Van Wyngaarden (josh.vanwyngaarden@gmail.com)
Brett F Dougherty (bbdougherty@gmail.com)
Bryan J Ladislas (bladislas@yahoo.com)
Gary L Helton (ghelton10@gmail.com)
Michael E Robinson (merobin@ufl.edu)
Samuel S Wu (samwu@biostat.ufl.edu)
Steven Z George (szgeorge@phhp.ufl.edu)

Version: 8 Date: 10 June 2011

Author's response to reviews:

- The trial registration number was added to the abstract in both the manuscript and the submission system
- The country was added to the affiliations details
- Vertical lines from the table were removed
- Figures were properly oriented and sized